Author’s response to reviews

Title: Camel Milk Ameliorates Steatohepatitis, Insulin Resistance and Lipid Peroxidation in Experimental Non Alcoholic Fatty Liver Disease

Authors:

Aida A Korish (iaidakorish@yahoo.com)
Maha M Arafah (marafah@hotmail.com)

Version: 2 Date: 3 April 2013

Author’s response to reviews:

Dear Editor in Chief of BMC Complementary & Alternative Medicine

It is my pleasure to have the chance to submit my manuscript titled:

“Effect of Camel Milk on Steatohepatitis, Insulin Resistance and Lipid Peroxidation in Rats on High Fat-Rich Cholesterol Diet”

In the current study we reported and for the first time that camel milk is effective in inhibition of the steatohepatitis, insulin resistance and lipid peroxidation triggered by HFD. This action would add an important strongly desired effect of camel milk in the prevention of fatty liver and insulin resistance associated with obesity. This in addition to the reported health promoting effects of CM is very promising and intensified the importance of this natural compound, with unknown major side effects, in achieving the goal of utilizing functional foods in treating one or more of the major health problems associated with the current unhealthy life style and feeding habits that leads to hyperlipidemia and insulin resistance with the consequent plethora of disorders constituting the metabolic syndrome

The content of the manuscript is original and it has not been published or accepted for publication, either in whole or in part, in any form. No part of the manuscript is currently under consideration for publication elsewhere, and no conflict of interest is present. All authors have contributed to the work and agreed to submit it for consideration to BMC Complementary & Alternative Medicine

Looking forward for acceptance of my manuscript for publication.

Corresponding author
Dr. Aida A Korish, MBBS, MSc, PhD
Associate Professor of Physiology
Faculty of Medicine
King Saud University
Phone: + (966)-506-282-704
Fax: + (966)-147-86798 or + (966)-146-72567
E-mail: iaidakorish@yahoo.com